Author's response to reviews

Title: Increased expression of integrin-linked kinase is associated with shorter survival in non-small cell lung cancer

Authors:

Iwao Takanami (takanami@med.teikyo-u.ac.jp)

Version: 4 Date: 28 December 2004

Author's response to reviews:

Exchanged highlights

Abstract is exchanged, including section heading.
Authors name is written exactly.
Tables The borders of tables should be visible black lines.
I performed the borders of tables are visible black lines. But my computer may have difficulty to make good line. I would be grateful to change my lines, if my arranged lines are not so good lines.
Figures- Figure numbers are changed.
Figure 1a-Figure 1
Figure 1b-Figure 2
Figure 1c-Figure 3
Figure 2-Figure 4
Additional files are deleted.